Author's response to reviews

Title: Health Literacy and Medical Care Costs in an Integrated Healthcare System: A cross-sectional cohort study

Authors:

Jolie N Haun (joliehaun@gmail.com)
Nitin Patel (nitin.patel4@va.gov)
Dustin D French (dustin.french@northwestern.edu)
Robert R Campbell (Robert.Campbell3@va.gov)
Douglas D Bradham (dbradham@air.org)
William Lapcevic (William.Lapcevic@va.gov)

Version: 2
Date: 9 February 2015

Author's response to reviews:

February 9, 2015

Jolie Haun, PhD EdS
8900 Grand Oak Circle (118M)
Tampa, FL, USA 33637
Phone: 904-377-6796
Email: joliehaun@gmail.com

To whom it may concern:

We are submitting the manuscript entitled, “Health Literacy and Medical Care Costs in an Integrated Healthcare System: A cross-sectional cohort study,” with minor changes requested upon submission.

1. Title page: We have provided email addresses of all authors.
2. Line numbering: We have included line and page numbers.
3. In compliance with editorial policies ethical approval information, e.g. institution and approval number, are included in the manuscript text. Data is reported according to reporting standards. The authors provide a review of data sources but require guidance if further detail is needed on data availability.

The authors of this article certify that the submitted manuscript, represents original work and that we have reviewed the final version and approve it for publication. Neither this manuscript nor a manuscript with substantially similar content under our authorship has been published or is currently being considered for publication by any other publication. This manuscript will not be submitted to any other publication while it is under consideration by BMC Health Services Research. We certify individuals listed as authors of this manuscript have participated in 1) substantial contributions to conception and design, or
acquisition of data, or analysis and interpretation of data; 2) drafting the article or revising it critically for important intellectual content; and 3) final approval of the version to be published. We certify that we have no financial affiliation/interest or potential conflicts of interest in the subject matter, materials, or products mentioned in this manuscript. Please feel free to contact me with any comments or questions.

Thank you for your consideration,

Jolie Haun, PhD EdS
Co-Core Director for Implementation/Dissemination
HSR&D Center of Innovation on Disability and Rehabilitation Research
James A. Haley Hospital